Author's response to reviews

Title: Ambulatory Medical Assistance - After Cancer (AMA-AC): A model for an early trajectory survivorship survey of lymphoma patients treated with anthracycline-based chemotherapy

Authors:

Gisele COMPACI (compaci.gisele@iuct-oncopole.fr)
Manuela RUETER (rueter.m@chu-toulouse.fr)
Sebastien LAMY (sebastien.lamy@univ-tlse3.fr)
Lucie OBERIC (oberic.lucie@iuct-oncopole.fr)
Christian RECHER (recher.christian@iuct-oncopole.fr)
Maryse LAPEYRE-MESTRE (maryse.lapeyre-mestre@univ-tlse3.fr)
Guy LAURENT (laurent.guy@iuct-oncopole.fr)
Fabien DESPAS (fabien.despas@univ-tlse3.fr)

Version: 8 Date: 12 October 2015

Author's response to reviews:

We would like to thank the editorial board for their helpful suggestions which have strengthened our manuscript. We have incorporated their suggestions into the revised version of the manuscript.

Editorial request:

Please state specifically in your manuscript whether the Ethical Committee of the Toulouse University Hospital approved your study.

The following sentence has been incorporated (line 155-116):
"The study has been approved by the ethical committee of the Toulouse University Hospital and all participants gave their written informed consent."

Sincerly yours
Dr. Fabien DESPAS